Author's response to reviews

Title: The impact of training non-physician clinicians in Malawi on maternal and perinatal mortality: a cluster randomised controlled evaluation of the Enhancing Training and Appropriate Technologies for Mothers and Babies in Africa (ETATMBA) project

Authors:

David Ellard (d.r.ellard@warwick.ac.uk)
Doug Simkiss (D.E.Simkiss@warwick.ac.uk)
Siobhan Quenby (S.Quenby@warwick.ac.uk)
David Davies (David.Davies@warwick.ac.uk)
Ngianga-bakwin Kandala (N-B.Kandala@warwick.ac.uk)
Francis Kamwendo (fwkamwendo@hotmail.com)
Chisale Mhango (cmhango@globemw.net)
Joseph Paul O-Hare (J.P.O-Hare@warwick.ac.uk)

Version: 2  Date: 10 October 2012

Author's response to reviews:

Added Acknowledgments section (Funder)